Author's response to reviews

Title: The Methodology for Developing a Prospective Meta-Analysis in the Family Planning Community

Authors:

David K Turok (david.turok@hsc.utah.edu)
Eve Espey (eespey@salud.unm.edu)
Alison B Edelman (edelmana@ohsu.edu)
Pamela S Lotke (plotke@email.arizona.edu)
Eva H Lathrop (elathro@emory.edu)
Stephanie B Teal (stephanie.teal@ucdenver.edu)
Janet C Jacobson (janet.jacobson@hsc.utah.edu)
Sara E Simonsen (sara.simenson@utah.edu)
Kenneth F Schulz (kschulz@fhi.org)

Version: 4 Date: 22 April 2011

Author's response to reviews:

This final version includes some minor grammatical changes and updates on the progress of all trials involved.